Author's response to reviews

Title: Patterns of Treatment Seeking Behavior for Mental Illnesses in Southwest Ethiopia: A Hospital Based Study

Authors:

Eshetu Girma (grm_sht@yahoo.com)
Markos Tesfaye (tesmarkos@yahoo.com)

Version: 4 Date: 16 August 2011

Author's response to reviews: see over
Response to Comments from the Associate Editor

Comment: I have one language related question though - the authors wrote “mentally ill him/herself could not respond” and I wanted to ask the authors if they can choose less stimatizing language such as "person with mental illness" rather than "mentally ill person"

Response: Thank you so much. It is a great comment. Now we have replaced all phrases with .....Person with mental illness......